Author's response to reviews

Title: Buccal Soft Tissue Lipoma In An Adult Nigerian: A Case Report and Literature Review.

Authors:

Adeyi A Adoga (adeyiadoga@gmail.com)
Tonga L Nimkur (nimkurtonga@yahoo.com)
Agabus N Manasseh (nanfwang@hotmail.com)
Godwin O Echejoh (oigechejoh@yahoo.com)

Version: 5 Date: 20 August 2008

Author's response to reviews:

Dear Editor,

We have gone through the third reviewer’s comments about the language corrections to be made on the manuscript. The reviewer did not state which ones are to be corrected. However, we have repeatedly gone through the manuscript again and made some appropriate language changes.

We hope our manuscript is duly considered.

Thank you.

Dr. Adeyi A. Adoga.